Author's response to reviews

Title: Effectiveness of different methods of health education: A comparative assessment in a scientific conference

Authors:

Asim Saha (asimsaha2311@yahoo.co.in)
Era Poddar (era12342000@yahoo.com)
Minal Mankad (minalmankad@rediffmail.com)

Version: 3 Date: 19 August 2005

Author's response to reviews:

Date 19th August 2005

To
The Editor
BMC Public Health

Respected Sir,

I have made necessary corrections in the format of the article ("Effectiveness of different methods of health education: A comparative assessment in a scientific conference" MS: 5858020164901775) as directed. Thanks for giving me a chance to publish our work in your esteemed journal. I hope now you will find this article in proper shape for publication.

Thanks again

Yours sincerely

Dr. Asim Saha,
Senior Research Officer (Medical),
Occupational Medicine Division,
National Institute of Occupational Health,
Meghani Nagar,
Ahmedabad-380 016
Gujarat, India.
E-mail: asimsaha2311@yahoo.co.in
Phone: +91(079) 22686351. Fax: +91(079) 22686110.